Author's response to reviews

Title: Treatment Strategies in Acute Traumatic Osseous Bankart Lesions Resulting from First Time Dislocation of the Shoulder A Two Year Follow-up Study

Authors:

Ulrich Spiegl (uli.spiegl@gmx.de)  
Christian Ryf (cryf@spitaldavos.ch)  
Pierre Hepp (Pierre.Hepp@medizin.uni-leipzig.de)  
Markus Loibl (markus.loibl@ukr.de)  
Paavo Rillmann (prillmann@spitaldavos.ch)

Version: 2 Date: 22 April 2013

Author's response to reviews: see over
Author’s response to Editorial Requirements

Title: Treatment Strategies in Acute Traumatic Osseous Bankart Lesions Resulting from First Time Dislocation of the Shoulder. A Two Year Follow-up Study

Authors:
Ulrich Spiegl (uli.spiegl@gmx.de)
Christian Ryf (cryf@spitaldavos.ch)
Pierre Hepp (Pierre.Hepp@medizin.uni-leipzig.de)
Paavo Rillmann (prillmann@spitaldavos.ch)

Version: 1 Date: 12 April 2013

Author's response to Editorial Requirements: See over
Research article: Treatment Strategies in Acute Traumatic Osseous Bankart Lesions Resulting from First Time Dislocation of the Shoulder A Two Year Follow-up Study

Authors: Ulrich Spiegl, Christian Ryf, Pierre Hepp, and Paavo Rillmann

In BMC Musculoskeletal Disorders

Editorial Requirements:

1.) Ethics statement:

- Included on page 5:
  An ethical approval was not necessary due to the retrospective study design. All patients signed an informed consent for participating in this study and for the publication of anonymous individual details.

2.) Please provide an "Authors' contributions" Section:

- Included on pages 9 and 10:
  Authors' contributions
  US participated in the study design, worked through the medical records, identified the patient collective, contacted all patients at the follow-up, performed the statistical analysis, and drafted the manuscript. CR performed a considerable amount of the operative procedures and helped in setting up the treatment strategy. PH participated in performing the statistical analysis, and helped to draft the manuscript considerably. PR conceived of the study, and participated in its design and coordination, performed a considerable amount of the operative procedures and helped to draft the manuscript.

3.) Please add "Acknowledgements" Section:

- Done on page 10:
  Acknowledgements
  We thank Thomas Perren and Frank Wurmitzer for participating in the clinical and surgical work of the patient collective and Markus Loibl for his support writing the manuscript.

- Permission obtained

- Sections adjusted to BMC Musculoskeletal Disorders on pages 9 to 13

4.) Kindly clarify whether patients provided written informed consent for the publication of individual details (as in table 1).

- Done on page 5:
  All patients signed an informed consent for participating in this study and for the publication of anonymous individual details.